Author's response to reviews

Title: An educational intervention to reduce pain and improve pain management for Malawian people living with HIV/AIDS and their family carers: study protocol for a randomised controlled trial

Authors:

Kennedy B Nkhoma (ntxkn1@nottingham.ac.uk)
Jane E Seymour (jane.seymour@nottingham.ac.uk)
Antony J Arthur (antony.arthur@uea.ac.uk)

Version: 3 Date: 2 July 2013

Author's response to reviews:

As requested by you on 2/7/13 we have now provided details of all authors' contributions including JS:

KN, JS and AA were responsible for the development and refinement of the protocol. KN is the principal investigator and trial manager. KN will conduct data analysis under the supervision of AA. KN and AA wrote the initial draft of the manuscript. KN, JS and AA contributed to, edited and approved the final manuscript.